RESEARCH ARTICLE

A retrospective cohort study of monthly rifampicin, ofloxacin and minocycline in the management of leprosy at the Hospital for Tropical Diseases, London, United Kingdom

Priyanka Sivakumaran^{1*}, Barbara de Barros^{2,3}, Vivianne Lopes Antonio Dias⁴, Diana N. Lockwood³, Stephen L. Walker^{2,3,5}

1 Royal London Hospital, London, United Kingdom, 2 Hospital for Tropical Diseases, University College London Hospitals NHS Foundation Trust, London, United Kingdom, 3 Faculty of Infectious and Tropical Diseases, London School of Hygiene and Tropical Medicine, London, United Kingdom, 4 Universidade Federal Fluminense, Rio de Janeiro, Brazil, 5 Department of Dermatology, University College London Hospitals NHS Foundation Trust, London, United Kingdom

* p.sivakumaran@nhs.net

Abstract

Introduction

The World Health Organization (WHO) recommends rifampicin, dapsone and clofazimine multi-drug therapy (MDT) for the treatment of leprosy. Severe adverse effects include dapsone hypersensitivity syndrome, skin pigmentation, haemolytic anaemia, and hepatitis. At the Hospital for Tropical Diseases (HTD), London, United Kingdom monthly rifampicin, ofloxacin and minocycline (mROM) is used as first line treatment for leprosy.

Objectives

To determine the clinical outcomes and experiences of individuals treated with mROM.

Methods

A retrospective study of individuals with leprosy who were prescribed mROM at HTD was conducted. Demographic and clinical data were collected on outcomes including relapses, leprosy reactions, bacterial index (BI) and adverse effects. Individuals were interviewed using a semi-structured questionnaire to understand their experiences of mROM.

Results

29 individuals were identified and 20 interviewed. 26 (89.7%) individuals completed monthly mROM. 9 (31%) had switched from WHO MDT to mROM (five of whom (55.6%) were interviewed). BI reduced significantly following mROM treatment (p = 0.04). 17 individuals (58.6%) experienced a leprosy reaction. One of the 29 (3.4%) relapsed. The relapse rate was 9.5/1000 person years. 49 reports of adverse effects were either mild or moderate. The most frequent adverse effect (14/49) reported was orange discolouration of urine. No



GOPEN ACCESS

Citation: Sivakumaran P, Barros Bd, Antonio Dias VL, Lockwood DN, Walker SL (2024) A retrospective cohort study of monthly rifampicin, ofloxacin and minocycline in the management of leprosy at the Hospital for Tropical Diseases, London, United Kingdom. PLoS Negl Trop Dis 18(12): e0012699. https://doi.org/10.1371/journal.pntd.0012699

Editor: Ben Pascoe, University of Oxford, UNITED KINGDOM OF GREAT BRITAIN AND NORTHERN IRELAND

Received: August 26, 2024

Accepted: November 14, 2024

Published: December 9, 2024

Copyright: © 2024 Sivakumaran et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: Payment of publication fees was supported by the Hospital for Tropical Diseases Charitable Fund which had no role in decision to publish, or preparation of the manuscript. **Competing interests:** The authors have declared that no competing interests exist.

adverse effect required hospitalisation or discontinuation of mROM. Most individuals reported that skin lesions improved by the time they had completed mROM.

Conclusions

In this small study in a non-endemic setting mROM was safe, effective and acceptable. mROM therapy is associated with improvement in skin lesions, decline in bacterial index and acceptable adverse effects. Larger, prospective, randomised studies are needed to determine whether relapse rates with mROM are equivalent or better than WHO MDT and to provide robust data on the seemingly better adverse effect profile of mROM.

Author summary

Leprosy is a chronic disabling infectious disease caused by *Mycobacterium leprae* and *Mycobacterium lepromatosis*. Anti-bacterial treatment with rifampicin, dapsone and clofazimine recommended by the World Health Organization (WHO) is effective but associated with significant adverse effects.

We report the effectiveness of monthly rifampicin, ofloxacin and minocycline in 29 individuals which was not associated with severe adverse effects. The monthly antimicrobial regime can be directly observed during routine clinical assessment reducing issues of adherence.

Alternative antibacterial regimes for leprosy, including newer anti-mycobacterial agents and monthly dosing, should be compared to the WHO recommended standard of care in well-designed randomised controlled trials to assess efficacy and safety.

Background

Leprosy is caused by *Mycobacterium leprae* and *Mycobacterium lepromatosis* and predominantly affects the skin and peripheral nerves [1]. Leprosy is a highly stigmatised disease due largely to the disability and permanent, visible physical impairments due to the neuropathy and immune-mediated inflammatory leprosy reactions. In 2022, 174 087 new cases of leprosy were reported to WHO by 128 countries and 95% of these cases were reported by 23 countries [2]. Leprosy is designated a neglected tropical disease (NTD).

The World Health Organization (WHO) recommends anti-microbial multi-drug therapy (MDT) to treat *M. leprae* and *M, lepromatosis* infection with three drugs: rifampicin, dapsone and clofazimine. WHO MDT is provided free of charge pre-packaged in 28-day blister packs (for adults and children older than 10 years) and distributed to governments by WHO. Each adult blister pack contains a monthly dose of rifampicin 600 mg, clofazimine 300 mg and dapsone 100 mg. Dapsone 100 mg and clofazimine 50mg are taken daily on the other 27 days. WHO recommends six months MDT for individuals with paucibacillary (PB) leprosy and 12 months for those with multibacillary (MB) leprosy [3].

WHO-recommended MDT is effective and was introduced in 1982 to standardise treatment and overcome the widespread problem of dapsone resistance following its use as monotherapy. MDT has high cure rates of 99% [4]. The reported relapse rate for participants taking 12 months MB MDT was reported to be 3.1% at 10 years in the Uniform Multidrug Therapy Regimen for Leprosy Patients (U-MDT) trial from Brazil [5,6]. It is estimated over 18 million people have been treated with MDT [7].

MDT is associated with adverse effects which can affect adherence, mental wellbeing and in some cases be fatal [8]. The Global Leprosy Programme has recognised the need for improved pharmacovigilance. In a study from Brazil 70.7% of adverse effects associated with MDT were attributed to dapsone [9]. Dapsone has dose-dependent adverse effects such as methaemoglobinaemia, bone marrow aplasia and haemolytic anaemia [8]. Dapsone Hypersensitivity Syndrome (DHS) has increased frequency in individuals with HLA-B*13: 01 genotype associated and is associated with significant mortality [10]. In a systematic study, a total of eleven prospective and retrospective studies were used to calculate an estimated incidence of DHS as 1.22% (range 0.82% to 3.0%) [8,10].

Clofazimine is a red dye and the commonest adverse effect is skin discoloration, ranging from red to black. The pigmentation usually fades within 12 months of stopping clofazimine, although traces of discoloration may remain for up to one year [11]. The skin discoloration associated with clofazimine is distressing, stigmatising and discloses the diagnosis to those who are aware of this adverse effect [12]. A study in India revealed 9.8% MB patients stopped taking their WHO-recommended MDT due to clofazimine pigmentation [13]. Clofazimine also causes a marked ichthyosis of the skin. A monthly dose of rifampicin as used in MDT causes a brief discolouration of body fluids for up to 48 hours after administration and very rarely hepatitis has been reported [13].

The recent WHO strategic framework for integrated control and management of skinrelated NTDs highlighted anti-microbial therapy for leprosy as a research gap in part because of the adverse effects associated with WHO-recommended MDT [14]. The current Global Leprosy Strategy 2021–2030 lists "More effective drugs or drug combinations, or shorter regimes, to treat leprosy" as a research area of "key importance" [7].

The Hospital for Tropical Diseases (HTD) in London operates the only dedicated Leprosy Clinic in the United Kingdom [15]. Leprosy in the UK is a rare disease and is often diagnosed late. There has been no reported autochthonous transmission since the 1954 [16]. Individuals with leprosy managed at the HTD were treated with WHO-recommended MDT unless adverse effects occurred in which case, they were switched to monthly rifampicin 600 mg, ofloxacin 400 mg and minocycline 100 mg (mROM). In 2016 it was decided to use mROM as first-line anti-microbial therapy for all individuals diagnosed with leprosy.

Rifampicin resistance may arise if clofazimine or dapsone are not taken regularly, a suggested alternative is mROM [17], which has been studied in a small number of controlled trials. In a sixth-month course of WHO MDT versus mROM in 268 individuals with paucibacillary (PB) leprosy in India, cure and relapse rates were similar in both groups [18]. A small study of 21 individuals with borderline lepromatous (BL) leprosy and lepromatous leprosy showed mROM for 24 months was as effective as MDT in clinical and bacteriological improvement [19]

We wished to determine the clinical outcomes and experiences of individuals treated with mROM at the HTD.

Methodology

Ethics statement

Ethical approval was granted by the London School of Hygiene and Tropical Medicine MSc Research Ethics Committee (27207) and the University College London Hospitals NHS Foundation Trust in accordance with the institutional policy.

Study design

We performed a retrospective cohort study to determine the clinical outcomes of individuals with leprosy treated with mROM at the HTD and any adverse effects attributed to mROM. Data were collected on the occurrence of leprosy reactions: Type 1 reactions (T1R) and ery-thema nodosum leprosum (ENL). T1Rs were treated with oral prednisolone. ENL was treated with oral prednisolone and/or thalidomide.

Individuals who were included in the retrospective analysis and still attending the Leprosy Clinic were invited to participate in a semi-structured interview about their experiences of taking anti-microbial medication including their recall of adverse effects. Translator services were used if needed.

The severity of reported adverse effects on activity of daily life was graded using the Common Terminology Criteria for Adverse Events [20].

Study setting

The HTD is a national referral centre in the UK for tropical and infectious diseases and has a dedicated Leprosy Clinic.

Data collection and study population

All individuals with a confirmed diagnosis of leprosy who attended the Leprosy Clinic and received mROM treatment between 1st January 2008 and 31st December 2021 were eligible. Data were collected from medical records in a standardised data collection form.

Our previously reported case definitions for the diagnosis of leprosy and leprosy reactions were used for the current study [21]. The diagnosis of leprosy and leprosy reaction was made based on the cardinal signs of leprosy or other clinical signs supported by histopathology.

The semi-structured questionnaire (S1 Interviewee questionnaire) was completed by a member of the research team (PS) as part of a face-to-face or telephone interview between 1st June-31st August 2022. All individuals who participated in the interviews gave written informed consent.

Data management and analysis

All data were anonymised and entered in Excel (S1 Data). Clinical data were analysed with descriptive statistics. Paired t-test was used to analyse change in bacterial index (BI). A p-value of <0.05 was considered statistically significant.

Adverse effects or symptoms reported by interviewees were attributed to the most likely cause based on the adverse effect profile of medications (individual components of mROM, prednisolone or thalidomide), the reported timing of onset and the management instituted. Where an adverse effect could be attributed to two or more components of mROM then mROM was listed as the most likely cause rather than individual components.

Results

29 individuals received mROM and 26 (89.7%) completed treatment with mROM. 20 individuals were interviewed.

The demographic and clinical data are summarised in <u>Table 1</u>. The majority of individuals were male (69%), and most individuals had positive slit-skin smears including 12 (41.3%) who had a mean BI of 4 or more. Leprosy reactions occurred in 17 (48.7%) of individuals.

Four individuals had hypertension, four had diabetes mellitus, and one had HIV infection. No individual was on a medication known to interact with rifampicin or ofloxacin or minocycline.

Characteristic		Number (%) [Total = 29]
Sex	Female	9 (31)
	Male	20 (69)
Age in years at diagnosis (median)	39 years (IQR: 31 to 50 years)	
Ridley-Jopling classification	TT	4 (13.8)
	BT	6 (20.7)
	BB	0 (0)
	BL	4 (13.8)
	LL	14 (48.3)
	PNL	1 (3.4)
Mean Bacterial Index at diagnosis	0	9 (31)
	0.1-0.99	0 (0)
	1-1.99	3 (10.3)
	2-2.99	1 (3.4)
	3-3.99	0 (0)
	4.4.99	3 (10.3)
	5-5.99	8 (27.6)
	6	1 (3.4)
	No data	4 (13.8)
WHO Disability Grade at diagnosis	0	9 (31)
	1	12 (41.3)
	2	8(27.6)
Anti-microbial treatment at diagnosis	WHO MDT	9 (31)
	mROM	20 (69)
Duration of mROM (months)	6	7 (24.1)
	12	13 (44.8)
	24	2 (6.9)
	Other	7 (24.1)
Follow-up status at time of data collection	Continues to attend clinic	23 (79.3)
	Discharged	4 (13.8)
	Deceased	2 (6.9)
Leprosy reactions experienced up to data collection	None	12 (41.3)
	Type 1 reaction	5 (17.2)
	ENL	9 (31)
	Both	3 (10.3)

TT, tuberculoid leprosy; BT, borderline tuberculoid; BB, mid-borderline; BL, borderline lepromatous; LL, lepromatous leprosy; PNL, pure neural leprosy; WHO MDT, World Health Organization multi-drug therapy; mROM, monthly rifampicin ofloxacin and minocycline; ENL, erythema nodosum leprosum.

https://doi.org/10.1371/journal.pntd.0012699.t001

Nine (31%) individuals were switched to mROM from WHO MDT. The reasons for the switch to mROM recorded in the patient records were as follows: five individuals were switched due to adverse effects of clofazimine, four for hyperpigmentation and one due to hyperpigmentation and ichthyosis. One individual stopped WHO MDT because of dapsone induced haemolytic anaemia and another due to dapsone supply issues. Two individuals did not adhere to daily treatment and so were offered mROM.

Five of the 29 patients were lost to follow-up. The total follow-up time was 96 years with a median of 3.10 years (IQR: 1.32 to 4.84 years). All individuals had a satisfactory response to

anti-microbial treatment. One individual relapsed two years after completing 12 doses of mROM (9.5/1000 person years).

Most individuals received a 6-month or 12-month course of mROM. Variations in duration was due to switching from WHO MDT and completing the course duration with mROM.

Twenty-five individuals had a baseline BI, the mean BI was 3 (SD 2.31). Nine had at least one repeat biopsy or smear. The mean time between the first BI and the last recorded BI was 39.2 months (SD: 21.0 months). Of these individuals the mean BI at diagnosis was 4.00 ± 1.94 and the mean of the last recorded BI was 2.69 ± 3.72 . This difference was statistically significant (p = 0.04).

Seventeen individuals had leprosy reactions (58.6%). Nine (52.9%) had ENL, five (29.5%) had T1R, three (17.6%) individuals had both T1R and ENL. All ENL and T1R required corticosteroids. Individuals with ENL were switched to thalidomide, 5 of the 17 individuals with reactions switched from WHO MDT. The odds of having a leprosy reaction at any time for those who received mROM compared to those who switched from WHO MDT to mROM was 1.2 (95% CI:0.25 to 5.9) p = 0.822

The mean time from completing monthly mROM to interviews being conducted was 2 years (IQR: 0 to 3 years). 18 of 20 (90%) interviewees reported 49 adverse effects (Table 2).

All adverse effects were mild or moderate based on Common Terminology Criteria for Adverse Events. Orange urine due to rifampicin was the most common reported adverse effect of mROM; 14 out of 49 (28.6%). Most adverse effects were non-specific and could be attributed to one or more of mROM, leprosy reactions, corticosteroids or thalidomide, and leprosy per se. mROM was thought to be the cause of a further 15 (30.6%) adverse effects. mROM and

Putative cause of adverse effect	Reported adverse effect	Number (%) [Total 49]
Rifampicin	Orange urine	14 (28.6)
mROM or leprosy reaction	Myalgia	4 (8.2)
	Joint pain or stiffness	4 (8.2)
mROM	Nausea	3 (6.1)
	Dizziness	2 (4.1)
	Breathlessness	1 (2)
	Bruising	1 (2)
	Candidiasis	1 (2)
	Dermatitis	1 (2)
	Headache	1 (2)
	Menorrhagia	1 (2)
	Night sweats	1 (2)
	Tinea corporis	1 (2)
	Vomiting	1 (2)
	Yellow sclera	1 (2)
mROM or leprosy reaction treatment	Low mood	2 (4.1)
Leprosy reaction treatment	Insomnia	2 (4.1)
	Loss of concentration	1 (2)
Leprosy reaction	Oedema	2 (4.1)
	Fever	1 (2)
Leprosy reaction or leprosy reaction treatment	Fatigue	1 (2)
Leprosy or leprosy reaction	Paraesthesia	2 (4.1)
Leprosy	Epistaxis	1 (2)

Table 2. Forty-nine adverse effects reported by 18 of 20 interviewees.

https://doi.org/10.1371/journal.pntd.0012699.t002

leprosy reaction or mROM and leprosy reaction treatment were thought to be responsible for 10 (20.4%) reported adverse effects. The remaining 10 (20.4%) were attributed to leprosy, leprosy reactions or leprosy reaction treatment. No adverse effects needed hospitalisation and no individual stopped mROM due to adverse effects. 50% of adverse effects occurred within 24 hours of starting mROM.

Discussion

We report good clinical outcome in a cohort of 29 individuals who received mROM for leprosy and were followed for a median period of 3.1 years. The proportion of individuals who experienced a leprosy reaction was 58.6% which is consistent with leprosy reaction rates of approximately 65% at three years in a large Brazilian cohort of individuals with MB leprosy treated with WHO MDT containing rifampicin, clofazimine and dapsone for six or 12 months [5]. Pre- and post-mROM treatment BI data were available in 31% and indicated a significant decline in BI. One individual relapsed following mROM. Interestingly this was their second relapse. mROM having been chosen to treat their lepromatous leprosy in part because they had relapsed following treatment with rifampicin and dapsone for borderline tuberculoid leprosy several years earlier.

Ten individuals (eight of whom had lepromatous leprosy) were successfully treated with 12 or 24 months of monthly rifampicin, moxifloxacin and minocycline in the United States [21]. One individual experienced transient mild elevation of liver transaminases and two had "minor gastrointestinal side effects". Another study by Faust et al. reported mROM used to treat three patients. Two with tuberculoid leprosy received six doses and one with BL leprosy received 24 doses. There were no adverse events requiring discontinuation of treatment [22].

All three anti-bacterial drugs in mROM are bactericidal for *M. leprae* whereas in WHO MDT dapsone is bacteriostatic and clofazimine only weakly bactericidal. A single dose of mROM was shown to exhibit \geq 97.5% bactericidal activity in nine of 10 individuals with lepromatous leprosy [17]. Mouse studies indicate that a single dose of mROM has less bactericidal activity than one month of WHO MDT [17] but the authors felt that it would be reasonable to compare several monthly doses of mROM with WHO MDT used for a similar duration.

Our cohort interviewees reported 49 potential adverse effects of which 79.6% were attributable to a component of mROM. No severe adverse effects were reported by the interviewees or documented in their medical records. A recent retrospective Brazilian study of the medical records of 443 individuals with leprosy showed that the risk of adverse effects was significantly greater in those who were treated with WHO MDT than those treated with mROM for 12 or 24 months [23]. Thirty-four of 247 (13.8%) individuals treated with mROM experienced an adverse reaction but there were no reports of rifampicin associated discolouration of urine which accounted for 35.9% of mROM related adverse effects in our interviewees.

Interestingly tendinopathies did not occur in our cohort nor the Brazilian study. Fluoroquinolones (including ofloxacin) are the subject of prescribing restrictions by regulatory agencies in the United States, United Kingdom and European Union [24–26] because of the increased risk of tendinopathy and other serious adverse effects including aortic dissection. A systematic review and meta-analysis by Alves et al. described an increased risk of tendon disorders from fifteen studies of individuals treated with fluoroquinolones [27]. Older age (≥ 60 years) and concomitant use of corticosteroids, particularly important in the context of leprosy reactions, were additional risk factors. Ofloxacin was associated with a higher rate of tendon disorders compared to ciprofloxacin and levofloxacin [28] in a more recent meta-analysis. Neither of these systematic reviews was able to comment in detail on the dose and duration of fluoroquinolone used but it would be reasonable to assume they were not used as a single monthly dose although a single dose of ofloxacin 400 mg is indicated for gonococcal urethritis and cervicitis due to susceptible *Neisseria gonorrhoeae* [29].

Our study has several limitations. The small number of participants and retrospective nature mean data should be interpreted with caution and the results of interviews often conducted months or years after completion of mROM may have been subject to recall bias and difficulty in attributing causality for adverse effects reliably. However, the use of interviews allows the voice of affected individuals to contribute to the understanding of the adverse effect profile of mROM which may not be captured by clinician documentation in medical records.

Our experience of the clinical effectiveness of mROM and the associated mild adverse effects has led us to use this combination first-line for the management of leprosy. The antimicrobials are usually taken during the monthly clinic attendance when nerve function assessment is undertaken. Treatment adherence is assured and "pill burden" reduced. Recently moxifloxacin has replaced ofloxacin because of the greater bactericidal activity of the former [30]. Robust clinical trials of alternative regimes to WHO MDT are required to provide evidence for efficacious (and potentially shorter) antibacterial regimes with reduced risk of severe adverse effects for people with leprosy. There is sufficient evidence to suggest that one intervention arm should be a monthly regime of rifampicin, moxifloxacin and minocycline.

Supporting information

S1 STROBE checklist. STROBE Statement—checklist of items that should be included in reports of observational studies. (DOC)

S1 Interviewee questionnaire. Blank questionnaire sheet used to collect data during patient interviews.

(DOCX)

S1 Data. Clinical data on 29 patients with leprosy. (XLSM)

Acknowledgments

We are grateful to the people living with leprosy who attend the Leprosy Clinic at the Hospital for Tropical Diseases, London for their participation.

Author Contributions

Conceptualization: Diana N. Lockwood, Stephen L. Walker.

Data curation: Priyanka Sivakumaran, Diana N. Lockwood, Stephen L. Walker.

Formal analysis: Priyanka Sivakumaran.

Investigation: Priyanka Sivakumaran, Barbara de Barros, Vivianne Lopes Antonio Dias, Diana N. Lockwood, Stephen L. Walker.

Methodology: Priyanka Sivakumaran, Stephen L. Walker.

Project administration: Stephen L. Walker.

Resources: Barbara de Barros, Stephen L. Walker.

Software: Barbara de Barros, Stephen L. Walker.

Supervision: Barbara de Barros, Diana N. Lockwood, Stephen L. Walker.

Validation: Stephen L. Walker.

Visualization: Stephen L. Walker.

- Writing original draft: Priyanka Sivakumaran, Barbara de Barros, Diana N. Lockwood, Stephen L. Walker.
- Writing review & editing: Priyanka Sivakumaran, Barbara de Barros, Diana N. Lockwood, Stephen L. Walker.

References

- 1. Chen KH, Lin CY, Bin Su S, Chen KT. Leprosy: A Review of Epidemiology, Clinical Diagnosis, and Management. J Trop Med. 2022;2022. https://doi.org/10.1155/2022/8652062 PMID: 35832335
- World Health Organization. Global leprosy (Hansen disease) update, 2022: new paradigm—control to elimination. Weekly epidemiological record. 2023; 409–430.
- 3. World Health Organization Regional Office for South-East Asia. Guidelines for the Diagnosis, Treatment and Prevention of Leprosy. New Delhi; 2017.
- Butlin CR, Jai K, Aung M, Withington S, Nicholls P, Alam K. Levels of disability and relapse in Bangladeshi MB leprosy cases, 10 years after treatment with 6m MB-MDT. Lepr Rev. 2019; 90: 388–398.
- Penna GO, Bührer-Sékula S, Kerr LRS, Stefani MM de A, Rodrigues LC, de Araújo MG, et al. Uniform multidrug therapy for leprosy patients in Brazil (U-MDT/CT-BR): Results of an open label, randomized and controlled clinical trial, among multibacillary patients. PLoS Negl Trop Dis. 2017; 11. https://doi.org/ 10.1371/journal.pntd.0005725 PMID: 28704363
- Penna GO, Pontes MA de A, Talhari S, Gonçalves H de S, Talhari C, Pessoa A de S, et al. Late relapses in leprosy patients in Brazil: 10-year post-trial of uniform multidrug therapy (U-MDT/CT-BR). Brazilian Journal of Infectious Diseases. 2024; 28. https://doi.org/10.1016/j.bjid.2024.103745 PMID: 38697216
- 7. World Health Organization. Towards Zero Leprosy.Global Leprosy (Hansen's disease) Strategy 2021–2030. New Delhi; 2017. licence:CCBY-NCSA3.0IGO
- Hilder R, Lockwood D. The adverse drug effects of dapsone therapy in leprosy: A systematic review. Lepr Rev. 2020; 91: 232–243. https://doi.org/10.47276/lr.91.3.232
- Bernardes Goulart IM, Arbex GL, Hubaide Carneiro M, Scalia Rodrigues M, Gadia R. Efeitos adversos da poliquimioterapia em pacientes com hanseníase: um levantamento de cinco anos em um Centro de Saúde da Universidade Federal de Uberlândia. Rev Soc Bras Med Trop. 2002; 35: 453–460.
- Zhang F-R, Liu H, Irwanto A, Fu X-A, Li Y, Yu G-Q, et al. HLA-B*13:01 and the Dapsone Hypersensitivity Syndrome. N Engl J Med. 2013; 369: 1620–1628. <u>https://doi.org/10.1056/NEJMoa1213096</u> PMID: 24152261
- 11. Moore VJ. A review of side-effects experienced by patients taking clofazimine. Lepr Rev. 1983; 54: 327–335. https://doi.org/10.5935/0305-7518.19830039 PMID: 6668986
- Nogueira AS, Garcia MAC, Silva MB da, Costa PF da, Frade MAC, Salgado CG, et al. Clofazimineinduced cutaneous hyperpigmentation as a source of stigma in the treatment of leprosy: A cross-sectional study. Tropical Medicine and International Health. 2024; 29: 327–333. https://doi.org/10.1111/tmi. 13977 PMID: 38348585
- 13. Mylan. Rifampicin 300mg capsules SmPC. 18 Oct 2024 [cited 6 Nov 2024]. Available: medicines.org.uk
- Ending the Neglect to Attain the Sustainable Development Goals: A Strategic Framework for Integrated Control and Management of Skin-Related Neglected Tropical Diseases. Geneva: World Health Organization; 2022. licence:CCBY-NC-SA3.0IGO
- Lockwood DN, McIntosh A, Armstrong M, Checkley AM, Walker SL, McBride A. Diagnosing and treating leprosy in a non-endemic setting in a national centre, London, United Kingdom 1995–2018. PLoS Negl Trop Dis. 2022; 16. https://doi.org/10.1371/journal.pntd.0010799 PMID: 36264976
- Gill AL, Gill G V., Beeching NJ. Familial transmission of leprosy in post-war Britain—Discrimination and dissent. QJM: An International Journal of Medicine. 2008; 101: 407–413. https://doi.org/10.1093/qjmed/ hcn030 PMID: 18375476
- Ji B, Sow S, Perani E, Lienhardt C, Diderot V, Grosset J. Bactericidal Activity of a Single-Dose Combination of Ofloxacin plus Minocycline, with or without Rifampin, against Mycobacterium leprae in Mice and in Lepromatous Patients. Antimicrob Agents Chemother. 1998; 42: 1115–1120. https://doi.org/10. 1128/AAC.42.5.1115 PMID: 9593137

- Kumar A, Girdhar A, Girdhar B. A randomized controlled trial to compare cure and relapse rate of paucibacillary multidrug therapy with monthly rifampicin, ofloxacin, and minocycline among paucibacillary leprosy patients in Agra District, India. Indian J Dermatol Venereol Leprol. 2015; 81: 356–362. https:// doi.org/10.4103/0378-6323.159929 PMID: 26144850
- Villahermosa LG, Fajardo TT, Abalos RM, Cellona R V, Balagon M V, Dela Cruz EC, et al. Parallel assessment of 24 monthly doses of rifampin, ofloxacin, and minocycline versus two years of World Health Organization multi-drug therapy for multi-bacillary leprosy. Am J Trop Med Hyg. 2004; 70: 197– 200. https://doi.org/10.4269/ajtmh.2004.70.197 PMID: 14993633
- U.S Department of Health and Human Services. Common Terminology Criteria for Adverse Events (CTCAE) Common Terminology Criteria for Adverse Events (CTCAE) v5.0. 2017 [cited 6 Nov 2024]. Available: ctep.cancer.gov
- Franco-Paredes C, Garcia-Creighton E, Henao-Martínez A, Kallgren DL, Banjade R, Dyer JA, et al. Novel approaches in the treatment of Hansen's disease (Leprosy): a case series of multidrug therapy of monthly rifampin, moxifloxacin, and minocycline (RMM) in the United States. Ther Adv Infect Dis. 2022; 9. https://doi.org/10.1177/20499361221135885 PMID: 36387060
- Faust L, Klowak M, Macrae C, Kopalakrishnan S, Showler AJ, Boggild AK. Ofloxacin-Containing Multidrug Therapy in Ambulatory Leprosy Patients: A Case Series. J Cutan Med Surg. 2021; 25: 45–52. https://doi.org/10.1177/1203475420952437 PMID: 32869655
- Celestino IC, Antunes DE, Santos DF, Gimenes VL, de Souza FM, Goulart IMB. Adverse reactions induced by MDT/WHO (Rifampicin+Clofazimine+Dapsone) and ROM (Rifampicin+Ofloxacin+Minocycline) regimens used in the treatment of leprosy: a cohort study in a National Reference Center in Brazil. Front Pharmacol. 2024; 15. https://doi.org/10.3389/fphar.2024.1346169 PMID: 38515839
- 24. FDA: In Brief: FDA warns that fluoroquinolone antibiotics can cause aortic aneurysm in certain patients. [cited 6 Nov 2024]. Available: https://www.fda.gov/news-events/fda-brief/fda-brief-fda-warns-fluoroquinolone-antibiotics-can-cause-aortic-aneurysm-certain-patients#:~:text=The%20U.S.%20Food%20and%20Drug,vessel%20blockages%20or%20aneurysms%2C%20high
- 25. Medicines and Healthcare products Regulatory Agency. Fluoroquinolone antibiotics: must only be prescribed when other commonly recommended antibiotics are inappropriate. 22 Jan 2024 [cited 6 Nov 2024]. Available: https://www.gov.uk/drug-safety-update/fluoroquinolone-antibiotics-must-now-onlybe-prescribed-when-other-commonly-recommended-antibiotics-are-inappropriate
- 26. European Medicines Agency. Fluoroquinolone and quinolone antibiotics: PRAC recommends restrictions on use New restrictions follow review of disabling and potentially long-lasting side effects. 2018 [cited 6 Nov 2024]. Available: https://www.ema.europa.eu/en/news/fluoroquinolone-antibiotics-reminder-measures-reduce-risk-long-lasting-disabling-potentially-irreversible-side-effects#:~:text=Fluoroquinolone%20treatment%20should%20be%20discontinued,of%20potentially%20irreversible% 20adverse%20reactions.
- Alves C, Mendes D, Marques FB. Fluoroquinolones and the risk of tendon injury: a systematic review and meta-analysis. Eur J Clin Pharmacol. 2019; 75: 1431–1443. <u>https://doi.org/10.1007/s00228-019-02713-1</u> PMID: 31270563
- Sangiorgio A, Sirone M, Adravanti FM, Testa EA, Riegger M, Filardo G. Achilles tendon complications of fluoroquinolone treatment: a molecule-stratified systematic review and meta-analysis. EFORT Open Rev. 2024; 9: 581–588. https://doi.org/10.1530/EOR-23-0181 PMID: 38949172
- 29. Mylan. Ofloxacin 200mg tablets SmPC. 28 Oct 2024 [cited 6 Nov 2024]. Available: medicines.org.uk
- **30.** Ji B, Grosset J. Combination of rifapentine-moxifloxacin minocycline (PMM) for the treatment of leprosy. Lepr Rev. 2000; 71: 81–87.